Author’s response to reviews

Title: Prevalence, impacts and medical managements of premenstrual syndrome among female students: cross-sectional study in college of health science, Mekelle university, Mekelle, northern Ethiopia

Authors:

Fikru W Tolossa (fikrewaf@gmail.com)
Mebratu L Bekele (mebleq@gmail.com)

Version: 4 Date: 28 October 2013

Author’s response to reviews:

The ethical approval was obtained from Ethics Review Committee of College of Health Sciences, Mekelle University, Mekelle, Ethiopia. Members of the committee were, 1. Afework Mulugeta, 2. Mesafint Molla, 3. Dagim Ali. The written consent was obtained from the study participants (College of Health Science, Mekelle University female students)